

Article



Barriers for Prosumers' Open Business Models: A Resource-Based View on Assets and Data-Sharing in Electricity Markets [†]

Mehdi Montakhabi ^{1,2,*}, Ine Van Zeeland ¹ and Pieter Ballon ¹

- ¹ imec-SMIT, Vrije Universiteit Brussel, Pleinlaan 9, 1050 Brussels, Belgium; ine.van.zeeland@vub.be (I.V.Z.); pieter.ballon@vub.be (P.B.)
- ² Center of Energy Technologies (CET), Department of Business Development and Technology (BTECH), Aarhus University, 8000 Aarhus, Denmark
- * Correspondence: mehdi.montakhabi@vub.be
- + This article is an extended version of our paper published in the proceedings of the 16th IEEE International Conference on Distributed Computing in Sensor Systems (DCOSS 2020) (pp. 482–489).

Abstract: This article explores the barriers for open business models in support of sustainability in electricity markets. It puts forward privacy and data protection concerns about sharing prosumers' physical assets as well as data due to to their critical role in decentralized modes of electricity/flexibility trading. In particular, it uses a multiple case study approach to identify actors' resources, examine other interested actors in each resource, define their objectives, and consider privacy and data protection concerns of sharing prosumers' physical assets and data. The findings yield new insights into sharing opportunities beyond electricity/flexibility trading. In doing so, our study contributes to theories of the firm by applying the resource-based view in a new context and to the business model literature by shedding light on barriers in applying open business models.

check for **updates**

Citation: Montakhabi, M.; Van Zeeland, I.; Ballon, P. Barriers for Prosumers' Open Business Models: A Resource-Based View on Assets and Data-Sharing in Electricity Markets. *Sustainability* 2022, *14*, 5705. https:// doi.org/10.3390/su14095705

Academic Editors: Elkhan Richard Sadik-Zada and Andrea Gatto

Received: 15 March 2022 Accepted: 4 May 2022 Published: 9 May 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). **Keywords:** circular business model; open business model; sharing economy; circular economy; peer-to-peer; electricity trading

1. Introduction

Adverse results of climate change which are increasingly manifesting around the globe [1] have made combating climate change an international mission [2]. The European Green Deal [3] and Paris Agreement [4] are two examples of international efforts to fight climate change. Energy transition as a path from "status quo to the envisioned future" [5] is said to be one of the most effective preventive acts against climate change [6].

While a considerable share of the extracted energy (from any source), before consumption, is transformed into electricity [7], the electricity industry is experiencing a rapid change toward more digital and decentralized forms of commercial interactions [8]. The electricity market is a highly regulated market that has been dominantly controlled by governments, with a clear division of labor and established roles [9]. In this situation, innovations do not frequently happen at the level of firms [10]. Nevertheless, digital and decentralized forms of commercial interactions are said to transform the industry from the bottom up [11]. Technological breakthroughs in the production of high-capacity batteries and solar panels at low prices, and the prevalence of smart devices, have facilitated the emergence of new market models (e.g., peer-to-peer, community self-consumption, and transactive energy models) [12]. Accordingly, electricity markets are witnessing the emergence of new types of multipolar innovation ecosystems around these models [13,14]. Not only do they innovate the paradigms of value generation and capturing, they also transcend firms' boundaries and transform a wide range of established organizations [15].

The wave of the 5D global energy megatrends, namely, decarbonization, decentralization, digital transformation, democratization, and disruption-as-usual, is said to have accelerated the shift from the conventional electricity paradigm to a new era of decentralized, distributed, clean, and smart energy systems [16]. The interactions between the urgent need to tackle climate change, advances in information and communication technologies (ICTs), and the proliferation of distributed energy resources, batteries, and home energy management systems, are, to a large extent, the main antecedents for this change [17]. New market models are said to have the capacity to generate a wide range of economic, social, and environmental values [14,18].

New market models in the electricity market have attracted scholars' and practitioners' attention over the past years [19,20]. Technical challenges of these models, such as their impacts on the power grid [21], different market and pricing mechanisms [22], security and data protection aspects [23], and (other) legal requirements [24] have been extensively studied, generating important insights toward understanding decentralized electricity/flexibility trading.

However, despite their potential value, other opportunities for sharing in the electricity market, beyond electricity and/or flexibility trading, have been overlooked in the literature [25]. Alongside the emergence of new market models, which are based on electricity/flexibility trading [26], the concept of sharing can potentially be applied beyond trading electricity or flexibility between peers [27]. This gap in the literature gains importance because some of these opportunities can already be exploited, whereas market models require several co-innovations and adoptions for emergence [28].

The purpose of this article is to explore the barriers for open business models [29] in support of sustainability in electricity markets. To do so, we narrow our focus to sharing phenomenon as a means for manifestation of open business models [30] in the electricity market and conduct a multiple case study [31] on the P2P, CSC, and TE models. The study builds upon [32]'s perspective on framing prosumers as rational market actors (i.e., firms) and therefore utilizes theories of the firm [33], resource-based theory [34] in particular, to explore opportunities for sharing physical assets as well as data. We winnow our discussions down to prosumers as one of the main emerging actors [35] in electricity markets due to their pivotal role in decentralized models. We investigate drawbacks in implementing prosumers' open business models from the perspective of privacy and data protection [36].

The findings yield insights into sharing opportunities beyond electricity/flexibility trading that have been overlooked so far. In doing so, our study contributes to theories of the firm by applying the resource-based view [37] in a new context (i.e., the electricity market) and to the business model literature by shedding light on prosumers' business models [38] through identifying barriers in applying open business models (i.e., by examining privacy and data protection concerns on sharing prosumers' physical assets and data).

This article extends previous research [39] (the preliminary results of this study were presented at the 16th IEEE International Conference on Distributed Computing in Sensor Systems (DCOSS 2020)) by conducting an extensive literature review on the theoretical and conceptual groundings of the study, improving the methodology by introducing detailed steps for data gathering and data analysis, providing a privacy discussion on the opportunities for sharing around prosumers' physical assets as well as their data, and providing discussions on prosumers' open business models [40]. This article adds two new contributions to our previous research [39] (see the second and third contributions in the next paragraph).

The novel contributions of this article are three-fold:

 First, it contributes to theory by applying resource-based theory to a relatively new context—electricity markets—to identify resources with sharing potential held by different existing and emerging actors. To our knowledge, resource-based theory has not been used before in the context of the electricity trading to discover sharing opportunities [41].

- Second, it introduces new insights on prosumers' business models [42] based on possibilities for sharing beyond trading surplus electricity from distributed resources, or prosumers' flexibility in future electricity markets [43].
- Third, it analyzes the differences between privacy [44] aspects of sharing physical objects versus data protection [45] issues related to sharing prosumers' data in electricity markets. We contend that physical privacy aspects are too often overlooked, though they can present critical barriers to prosumer sharing.

The rest of this manuscript is organized as follows. Section 2 provides a literature review on new market models for electricity and/or flexibility trading, open business models for circularity, and theories of the firm. This is followed by the methodology section, which introduces boundary conditions, case study selection, data gathering, and data analysis (Section 3). Section 4 presents the traditional electricity trading paradigm and introduces three cases: P2P, CSC, and TE. Section 5 presents the findings of the study. Valuable resources of the different actors in the electricity market with a potential for sharing, interested actors, and the benefits they can gain by accessing each resource, are presented in this section. Consequently, opportunities for sharing in the electricity market are introduced. Section 6 analyzes the findings. First, the current status of sharing is compared with future opportunities. Secony, privacy and data protection aspects of prosumers' sharing opportunities are analyzed. Section 7 critically discusses the findings and compares the results with others. Furthermore, we outline how our study contributes to theory. Section 8 discusses the limitations of this study and opportunities for future research. Section 9 concludes the paper.

2. Literature Review

2.1. New Market Models for Electricity or Flexibility Trading

When it comes to new market models for electricity trading, many initiatives can be detected, including peer-to-peer (P2P) electricity trading, community self-consumption (CSC), and transactive energy (TE) models [38]. There is not yet a clear distinction between the three market models, but they promise similar benefits. Although the three models have their own merits, there is no common understanding about them. Knowing that they may be self-contradictory in some situations, it becomes important to highlight the alignments and contradictions between them [22].

There is a rich literature on various market models for electricity transactions. The most cited models are the P2P, the CSC, the TE, or a combination of them [22]. A common element of these models is the encouragement of and incentives for prosumers (and consumers) to take up an active role in electricity markets. This element is usually achieved by allowing prosumers to trade electricity/flexibility with other market participants, such as other prosumers and grid operators, in return for some (financial) incentives. However, the models also have their own specific objectives [32].

2.2. Open Business Models for Circularity

Companies are increasingly exploring and exploiting external resources and knowledge by increasing their openness to innovating their business models, in order to gain a competitive advantage [46–48]. Yet, research on open business models remains scarce [49]. In our study, we define open business models as "a subclass of business models in which collaboration of the focal firm with its ecosystem is a decisive or novel element of value creation and capturing" [49] (p. 175). Through open business models, a company becomes part of a larger innovation ecosystem consisting of individuals, communities, and other organizations, which entails simultaneous competition and cooperation between ecosystem actors [50,51]. Thus, open business models enable an organization to be more effective in both creating and capturing value [52] by leveraging more ideas through inclusion of a variety of external concepts. At the same time, open business models allow greater value capture by utilizing a firm's key asset, resource, or position, not only in that organization's own operations, but also in other companies' businesses [40,52]. This resonates very well with the ideas of the circular and sharing economy and circular business models. In understanding why companies engage in open business models, ref. [49] identified five main antecedents that lead companies to open up their business models: (1) business model inconsistency, (2) a need to create and capture new value, (3) previous experience with collaboration, (4) open business model patterns, and (5) industry convergence. Thus,

core of open business models for circularity. The concept of the circular economy has gained attention in the literature by transforming the way resources are applied: by shifting from existing open production systems to closed production systems, where resources are reused and kept in a loop of production and consumption [53]. Thus, the circular economy provides large potential for sustainability transformation [54]. Yet, very little is known about the realization of the political goals and scientific principles attached to a circular economy in business practice [55]. Researchers have recognized that in order to facilitate a circular economy, fundamental transitions of established business strategies, value chains, and eventually business models are needed [56]. Hence, the circular economy may be described as a cyclic system that aims to eliminate waste by turning goods that are at the end of their life cycle into resources for new goods [57]. Closing material loops in industrial ecosystems can create a continual use of resources, which is achieved through long-lasting design, proactive maintenance, recycling, repairing, refurbishment, and re-manufacturing [58].

sharing opportunities for new circular business to create and capture new value are at the

The concept of circular business models has gained increasing focus (e.g., [59–61]) and may be defined as creating, delivering, and capturing value, while implementing circular strategies that can prolong the useful life of products and parts and close material loops [62], p. 187. A circular business should be built on distributed marketplaces or decentralized networks that create a sense of belonging, collective accountability, and mutual benefit through the community they build [63]. The market relations between the circular businesses differ between peer-to-peer (P2P), business-to-consumer (B2C), and business-to-business (B2B), where ref. [64] emphasizes sharing among peers as the most innovative and most interesting variant of the sharing economy. Increasingly, the sharing of products (as a service), specifically through digital sharing platforms, is seen as an enabler for a circular economy (e.g., [65]). Unlocking the potential of the circular economy depends on innovative large and complex dynamic data collection and analysis [66]. Thus, sharing data and opportunities with stakeholders is a critical dimension [67] as the circular economy depends on developing new business models.

2.3. Theories of the Firm

Theories of the firm are applied to state why firms exist and how they make decisions to maximize profits, compete, etc. In other words, they are applied to predict behaviors of firms. For more than four decades, economists, sociologists, and organizational scholars have extensively examined the theory of the firm's central question: What determines the boundaries of the firm? Many alternative theories have emerged and are frequently positioned as competing explanations, often with no shortage of critique for one another [68]. By building our study on [32]'s perspective, which considered prosumers as firms, we take the liberty of applying theories of the firm to discuss prosumers' open business models. In the following we provide an overview of three (out of many) theories of the firm—industrial organization theory, resource-based, and dynamic capabilities—that are mostly applied to tackle the increasing importance of transcending organizational boundaries through open business models in the literature [29]. Furthermore, we argue for choosing resource-based theory for this research.

Industrial organization theory considers the strategies that an organization would devise as a means to relate the firm to its environment. In this view, an organization selects a position that provides the best competitive conditions. An organization may decide to maintain this position or, if required, impact rivalry for its own benefit. As an example, we can mention strategies that hinder any new entry into the market in which the organization is competing. In this view, external events dictate a firm's strategies [69].

By contrast, the resource-based view takes an internal approach and considers strategies of organizations beyond responses to the actions of market forces. It prioritizes the resource base of an organization. Opportunities and threats (which are out of the control of organizations) are not the only identifiers of organizations' strategies [70]. Whether an organization continues to exist or maintain its superiority depends on whether it builds idiosyncratic capabilities: capabilities that distinguish it from competitors in changing markets [71]. In [70]'s view, industrial organization theory and resource-based theory complement each other, and can therefore be combined in the theory of industrial organization and resources.

Everything that a firm uses to achieve its planned goals is considered to be part of the firm's "resource base" [72]. Resources thus are "all assets, capabilities, organizational processes, firm attributes, information, knowledge, etc. controlled by a firm that enable the firm to conceive of and implement strategies that improve its efficiency and effective-ness" [73], p. 101. Resource-based theory considers specific types of firms' resources (i.e., VRIN resources) as potential basis for building competitive advantages.

Firms in competition possess various types of resources, which are mostly immobile. The immobility of firms' resources intensifies the variety of their competitive positions in the long run [74,75]. Value, rarity, inimitability, and non-substitutability are the characteristics of special resources of organizations, which impact their market selection and increase their success rate [76]. The possession of special resources is required for the success of a firm. Furthermore, possession of distinguishing capabilities enables a firm to effectively exploit its resources [75].

Resource-based theory has been criticized for being too static and for ignoring the dynamism of markets in recent decades, when competition was fiercer and markets were more turbulent [77]. Resource-based theory assumes that (a) resources are heterogeneously persistent, and (b) benefits originating in a lack of competition in gaining complementary resources are long lasting [74]. Both assumptions lead to a failure of resource-based theory in explaining the effects of turbulent markets and the evolution of firms. Despite the effective influence of resource-based theory on strategic management, its validity is disputed as it cannot answer questions about turbulent market situations, nor transformational mechanisms that use resource-based theory into question because it fails to identify the mechanisms which allow firms to evolve their resources and capabilities, and transform their resources into competitive advantages.

Subsequently, the concept of "dynamic capabilities" can complement resource-based theory by considering resources' developing characteristics as well as developing capabilities of firms [80]. Dynamic capabilities are defined as the ability to build, develop, or amend the resource base of a firm. They are capabilities that are simultaneously part of the resource base of a firm. Dynamic capabilities can amend and develop themselves; they have the capacity to self-modify [72]. The concept of dynamic capabilities also completes the shortcomings of resource-based theory by justifying the effects of market dynamism [81]. Therefore, the relationship between the concept of dynamic capabilities and resource-based theory is complementary.

The concept of dynamic capabilities has been effectively used as a theoretical framework in abundant empirical research over the past two decades. It has been continually improved and now completes resource-based theory by filling its gaps, specifically to address the natural evolution of organizations' resources and capabilities under conditions of market turbulence. The concept of dynamic capabilities makes it possible to identify processes that are necessary for the evolution of the resources of firms, whether they are industry- or firm-specific [78].

Considering the purpose of this research, which is to identify sharing opportunities and consequently to identify barriers to apply prosumers' open business models, we opt for the resource-based view [37] for two main reasons. First, the sharing concept is tied with resources of others that can be deployed by the firm or resources of a firm that could be deployed by others [82]. Among the three discussed theories, the resource-based view is a better fit with the sharing concept. Second, the electricity market is, arguably, not a really turbulent market [83]. Therefore, between resource-based and dynamic capabilities, the former is a simpler theory to apply in this context.

3. Methodology

To carry out our study, an exploratory approach [84] was deployed. The overarching method for data gathering is multiple case study [31]. In the following subsections, boundary conditions, case study selection, data collection, and analysis are explained in detail.

3.1. Boundary Conditions

In this study, we apply the resource-based view [37]. Hence, each actor is considered as a bundle of resources under a common governance [85]. The resource base of a firm can have various components. Among the common categories of resources are:

- 1. Physical assets: grid infrastructure, smart meters, batteries, electric vehicles, etc.
- 2. Digital resources (data): smart meter information, flexibility-related information, supply and demand information, etc.

The resource base of a firm is not limited to these two categories. Other categories of resources have also been mentioned in literature, such as human resources, financial resources, downstream and upstream knowledge, governance-related (administrative) knowledge, and reputation-related resources, to name a few [85].

The sharing economy is concerned with non-VRIN resources [86]. Therefore, we use the two categories mentioned above, physical assets and data, to categorize the non-VRIN resources of different actors in the electricity market to which the sharing concept would apply.

3.2. Case Study Selection

As there is a flurry of initiatives centered on decentralized production and digitally enabled forms of transactions, we decided on a two-step process to narrow the focus. First, the empirical study was set to three cases in order to investigate actors and their resources as well as their motivations. Second, three controversial cases were selected yet with considerable similarities, that is, based on a decentralized nature, and with a diverse range of capacities in value generation [32] that affect electricity trading in similar ways, but at different capacities. The selected cases, arguably, are said to alternate the traditional electricity trading [87].

In doing so, setting market models as cases—thus, P2P, CSC, and TE models—they are not required to be limited to a specific project, pilot test, geographic region, or study [88], allowing a broad range of data for each case. While these are three different decentralized forms of electricity/flexibility trading changing electricity trading in different ways, the conclusions drawn are comparable.

3.3. Data Collection

In the first step of our study, extensive desk research [89] yielded insights into the identification of different existing and emerging actors in the electricity market. In particular, actors' resources, the objectives they pursue in the market, and other interested actors and their motivations are identified. This resulted in an overview of several actors in different market models (traditional, P2P, CSC, and TE models).

Subsequently, semi-structured interviews [90] were planned to validate the findings from the literature review (Appendix A shows the guideline questions for semi-structured interviews). For this purpose, the interviewees represent several stakeholder groups (e.g., existing actors, prosumers, policy makers, and academics) to provide a comprehensive view of the electricity market. Interview questions covered a wide range of aspects related to the current and future electricity markets (actors in the market, their responsibilities, resources, objectives, etc.). Top levels in management hierarchies were chosen for interviews to have a broad view on their companies' business models as well as a good understanding of the electricity market. To identify and select the interviewees, a snowball technique was deployed [13,91]. This helps to approach relevant people that otherwise tend to remain "under the radar"; this continues until data saturation is reached. A total of 26 interviews were conducted between October 2019 and March 2020. Interviews were conducted face-to-face and via Skype. Each interview took forty-five minutes on average.

3.4. Data Analysis

We transcribed interviews, coded the data, synthesized, built narratives, and applied storytelling techniques [92] to present the findings. Furthermore, in order to identify opportunities for sharing in the electricity market, a matrix was built on the findings of the study, which has actors in the electricity markets on its axes. For this study, we consider an opportunity to be a situation in which new services, products, raw materials, markets, and in general business models emerge through the formation of new means-ends combinations [93]. In this article, a sharing opportunity is a combination of a valuable resource of an actor and another actor interested in gaining access to that resource. The interested actor pursues one of several benefits by accessing the resource. In other words, the resource may be utilized for value generation or capturing by the interested actor.

The results of the matrix analysis were validated (i.e., face validity) by two expert members of the Global Observatory on Peer-to-Peer, Community Self-Consumption and Transactive Energy Models. We made a database of expert members in different five subtasks of Global Observatory: (i) power systems integration, (ii) hardware, software and data, (iii) transactions and markets, (iv) economic and social value, and (v) policy and regulation. Considering our research design, for the purpose of face validity, we opted for experts from sub-tasks ii and iii.

The next step in our analysis was to focus on the sharing opportunities of prosumers, and examine aspects of privacy and data protection related to the prosumer resources. To this end, we disambiguated concerns over privacy from concerns over personal data, based on a seminal study providing a comprehensive overview of different privacy typologies [94]. Applying these different types of privacy conceptions to the listed prosumer resources serves to clarify that sharing of physical assets can raise other privacy concerns (and merit a different type of protection) than merely concerns related to sharing data. Addressing these concerns will be instrumental to promoting prosumer sharing in electricity markets. Figure 1 shows a summary of our applied methods.



Figure 1. An overview of the applied methods.

4. Three Cases

The traditional electricity trading model refers to central electricity production from non-renewable (fossil-fuel, nuclear, gas, etc.) and renewable (wind, hydroelectric, etc.) resources in power plants. Electricity, in a one-directional flow, passes through the transmission grid, is transformed from high to low voltage, and is delivered to consumers through distribution grids [95]. Figure 2 shows the evolution of power systems during the time. In the following subsections, three cases (P2P, CSC, and TE) are introduced.



Figure 2. Evolution of power systems [14].

4.1. Case 1: Peer-to-Peer Electricity Trading

P2P market models support trading of electricity between prosumers (directly or through an intermediary) [39]. A possible future scenario of P2P energy trading is shown in Figure 3. Apart from prosumers, the scenario also envisions the presence of representatives who can trade electricity on behalf of citizens and a broker who facilitates and clears the market [35].



Figure 3. Peer-to-Peer electricity trading [13].

P2P models enable mutual transactions among different entities to trade electricity [96,97]. Energy traders in the P2P market may be of different sizes, i.e., residential houses, neighborhoods, microgrids, or local distribution networks [96,98]. These models are described as ways to allow the grid to take advantage of demand-side, flexible resources, operationally as well as economically [99]. In P2P models, the objective of the market mechanism is to incentivize transactions that prioritize maximizing the benefits of individual prosumers [100]. Such models could involve intermediate parties that facilitate the trades among prosumers, or support fully decentralized trades among them. The market mechanisms used in P2P models are usually set to optimize the trading based on algorithms with objectives of matching the excess supply of prosumers with the demand of consumers.

P2P electricity trading schemes enable consumers and prosumers in the same region to trade electricity they have produced with their own renewable energy resources among each other [101]. The type of sharing that occurs in these schemes is similar to what happens in most other sharing concepts that are based on P2P interaction [102–104], such as Airbnb and Uber.

4.2. Case 2: Community Self-Consumption

Community Self-Consumption (CSC) market models support communities in reducing their dependence from the electricity grid [105] through collectively utilizing their members' available resources [106]. Apart from aiming at grid independence, they could also provide flexibility services to other communities or to grid operators. Typically, a community manager orchestrates the market and assists members to collectively decides how to utilize their resources [107]. The assistance could range from having full responsibility for making decisions on behalf of the community members to only playing the role of a facilitator and strictly following the instructions given by the community's members. The community manager, the rights and privileges he or she holds, the scope of operation he or she has, and the length of his or her term are chosen by the community members via consensus mechanisms. Once these choices are made, the community manager's responsibility is to ensure that every community member receives a fair share of the rewards received according to their contribution to the service provided. The distribution of rewards is typically done via transactions between community members, between the community manager and community members, and the community manager and other communities/grid operators.

In CSC models, the overarching goal is independence from centralized electricity generation through unification of dispersed resources in communities [105]. Community members operate in a collaborative manner to optimize usage of resources [14,107]. CSC incentivizes transactions that prioritize maximizing the benefits of the community. CSC models usually involve a community manager [107] who coordinates transactions within the community as well as the transactions with other communities or the main grid. The market mechanisms used in CSC models are set to optimize the trading based on algorithms with objectives such as minimizing the electricity import by the community from the main grid and maximizing the revenue for the community. They usually involve sharing the individual prosumers' assets among each other or aggregating all the assets within the community in order to maximize the benefits for the entire community.

4.3. Case 3: Transactive Energy Models

Transactive Energy (TE) market models support grid operators in balancing the grid at all times [108]. They achieve this by facilitating demand response services provided by prosumers based on market-based incentives [109]. The key difference between the TE and P2P/CSC market models is that in TE models there is typically a single buyer—the grid operator or aggregator—who demands a certain amount of flexibility, thus offering financial incentives to prosumers to engage and provide flexibility. Prosumers are competing with each other on who would be selected for the service provision. Typically, the flexibility demanded is substantially larger than the flexibility an individual prosumer could provide [110]. Hence, typically, a number of prosumers is selected such that their aggregated flexibility satisfies the needs of the aggregator/grid operator. Similarly to CSC market models, all selected prosumers are rewarded in proportion to their contributed flexibility.

TE models are based on demand response, where end-user loads are automated and engaged through market-based interactions [108,110]. They provide market access to flexibility providers, and a support tool for the grid operators to manage technical complications [111]. It is a distributed control strategy that uses market mechanisms to engage self-interested responsive loads to provide services to the grid [112]. The objective of the market mechanism is to incentive transactions that prioritize and support the stability and reliability of the grid [108]. As the grid operators are the main actors who are responsible for maintaining the stability of the grid, they are usually the primary participants in the market buying electricity/flexibility directly from prosumers or via aggregators [113]. The

10 of 29

market mechanisms used in TE models are set to optimize trading based on algorithms with different objectives, such as keeping the grid in balance, reducing grid congestion, and maintaining the voltage and frequency stability.

5. Findings

5.1. Actors' Resources and Interested Parties

In this section, actors (existing and emerging) and their market objectives in the electricity market are briefly introduced. Resources of each of the actors are identified. For the list of actors in the future electricity market, the study considers the future scenarios introduced in [114] and actors introduced based on the future scenarios in [35]. Resources of each actor are categorized under physical and digital groups. An overview of these resources is given in Table 1.

Table 1. Valuable resources of each actor in the future electricity markets and interested actors in each resource.

	Physical Resources	Digital Resources (Data)—Interested Parties
Prosumer (Pro)	$\begin{array}{l} \operatorname{RES} \leftarrow \operatorname{Pro}, \operatorname{Rep} \\ \operatorname{Home} \text{ Battery} \leftarrow \operatorname{Pro}, \operatorname{Rep} \\ \operatorname{EV} \leftarrow \operatorname{Pro}, \operatorname{Rep} \\ \operatorname{EV} \text{ charging station} \leftarrow \operatorname{Pro}, \operatorname{Rep}, \operatorname{DSO} \\ \end{array}$ $\begin{array}{l} \operatorname{EV} \text{ battery} \leftarrow \operatorname{Pro}, \operatorname{Rep} \\ \operatorname{HEMS} \leftarrow \operatorname{Pro}, \operatorname{Rep} \\ \operatorname{Smart} \operatorname{Appliance} \leftarrow \end{array}$	Smart Meters \leftarrow Rep, Ret, DSO, TSO Demand \leftarrow Rep, Br, Ret, DSO, TSO Supply \leftarrow Rep, Br, Ret, DSO, TSO Flexibility \leftarrow Rep, DSO, TSO
Representative (Rep)	Clients' RES \leftarrow Pro, Rep Clients' Batteries \leftarrow Pro, Rep Clients' EVs \leftarrow Pro, Rep Clients' EV Charging Stations \leftarrow Pro, Rep, DSO Clients' Batteries of EVs \leftarrow Pro, Rep Clients' HMS \leftarrow Pro, Rep	Clients' Smart Meters \leftarrow Ret, DSO, TSO Clients Demand \leftarrow Br, Ret, DSO, TSO Clients' Supply \leftarrow Br, Ret, DSO, TSO Clients' Flexibility \leftarrow DSO, TSO
Broker (Br)		Sellers'supply information \leftarrow Ret, DSO, TSO Sellers'offered price \leftarrow Ret Buyers'demand information \leftarrow Ret, DSO, TSO Buyers offered price \leftarrow Ret Clearance price \leftarrow Ret Total traded volume \leftarrow Ret, DSO, TSO, Gen
Aggregator (Agg)		Clients' supply capacity \leftarrow TSO Clients' demand capacity \leftarrow TSO Clients' balancing capacity \leftarrow TSO
Retailer (Ret)		Customers' demand information \leftarrow DSO, TSO Customers' supply information \leftarrow DSO, TSO
DSO	Distribution grid infrastructure \leftarrow Pro, Rep, Ret	Smart meters' inflow information \leftarrow Rep, Ret Smart meters' outflow information \leftarrow Rep, Ret Congestion information \leftarrow Rep, Ret
TSO	Transmission grid Infrastructure \leftarrow TSO, Gen, Pro	Balancing information \leftarrow Agg, Ret, TSO, Gen Congestion information \leftarrow Agg, TSO, Gen Demand/supply pred. \leftarrow Agg, Ret, DSO, TSO, Gen
Generator (Gen)		Power plants (coal, gas, nuclear, etc.) \leftarrow Ret

Note: If the interested actor in a resource is the same as the owner, it indicates other peer actors are interested in that resource.

5.1.1. Prosumers

Prosumers are consumers who can also act as producers. In the electricity market, this means that they can generate electricity and inject it into the grid. To this end, they are in possession of renewable energy sources (e.g., solar panels) and storage devices (e.g., batteries). They will most likely also own smart meters, home energy management systems, and various other smart appliances. Prosumers' main objectives will be to *minimize cost* (specifically, their electricity bills), *maximize profit*, and *mitigate their dependency on the electricity grid*, by maximizing how they use their resources. To summarize, prosumers have the following valuable resources:

- A **renewable energy source (RES)** is a mini-generator located on a prosumer's premises (e.g., a solar panel). Typically, most of the electricity generated by a RES is consumed by its owner, who may inject surplus electricity into the grid.
- Home batteries are storage appliances that allow for intentional latency between the provision and the consumption of electricity generated or purchased by prosumers.
- A **smart meter** is an advanced measuring and recording device that keeps track of the electricity flowing in both directions (from the home to the grid, and vice versa) and that can perform two-way communications with other actors or appliances.
- A home energy management system (HMS) is a platform that consists of hardware and software to monitor electricity consumption and production. It allows prosumers to manually control and/or automate their household energy consumption.
- Demand information is information about the electricity requirements of prosumers, as well as their energy consumption patterns.
- Supply information is information about the volume of electricity produced by prosumers' RES, as well as production patterns.
- Flexibility information is information about the extent to which prosumers can modify their electricity production or consumption in response to variability, expected or otherwise.
- Smart appliances are Internet-connected appliances that connect to each other and/or other intelligent devices in the home. They can often be accessed and controlled remotely.
- An **electric vehicle (EV)** is a vehicle that has an electric motor for propulsion (or two such motors). EV owners typically also possess EV batteries and charging stations.
- An EV battery is a battery installed in an EV, which stores (and transports) electricity.
- An **EV charging station** is an appliance that can connect EVs to the grid to (dis)charge electricity.

5.1.2. Representatives (A New Role in Future Electricity Markets)

Representatives are a new type of actor in the peer-to-peer electricity market. Their role is to manage their clients' information and physical assets (RES, home battery, smart appliances, etc.) and to represent them in the peer-to-peer sharing market [35,114]. In doing so, they transform consumers' passive roles into active market participation. The presence of representatives can therefore increase consumer involvement in peer-to-peer electricity markets.

Representatives are expected to facilitate sharing opportunities between prosumers and other, already established market players.

Their main objectives will be to *minimize costs* and *increase profits* for their clients (prosumers). As representatives do not own their clients' assets, their resources mostly fall into categories other than physical and digital resources. Instead, representatives may for instance possess upstream knowledge (sourcing knowledge, and knowledge about product/service- or process-related technologies), downstream knowledge (which is critical to customer interfaces and is related to marketing, sales, distribution, and after-sales services), administrative (governance-related) resources, and reputable resources (brand names, a good reputation for honest business dealings, etc.).

5.1.3. Brokers (A New Role in Future Electricity Markets)

Like representatives, brokers take up a new role in future electricity markets [35]. They are intermediate, (neutral) actors facilitating the trade between peers. Due to their position, they have access to information about all parties participating in peer-to-peer electricity trading, as well as all transactions. Their objectives are to *clear the peer-to-peer market* while *respecting the grid's constraints*, as well as the *prosumers' preferences*.

As for representatives, brokers' special market position is not derived from their own physical or digital resources. Nor do they have upstream or downstream knowledge, or brand and other common resources. Brokers' positions mostly result from legal monopolies or first-mover advantages, which can provide actors with an important share of the market. Essentially, they will have access to all the information contained in a prosumer bid, such as the amount of electricity offered, the price requested, the demand and supply bids, buyer preferences (specific type of energy source, location, etc.), grid access points (location), and electricity source (solar, wind, biomass, etc.).

5.1.4. Aggregators

Aggregators are actors that exist already in current markets. They provide ancillary services to grid operators. They play an important role in balancing the electricity market by aggregating prosumers' flexibility and trading. Their main objective is to *maximize profit* by offering ancillary services to grid operators. To achieve this, they have access to the following information:

- Clients' supply capacity: Information about the capacity of their clients' electricity supply to the grid, including supply patterns.
- Clients' demand capacity: Information about the capacity of their clients' electricity demand from the grid, including demand patterns.
- Clients' balancing capacity: Information about the flexibility of their clients, which can be served to the grid.

5.1.5. Retailers

Retailers provide consumers with electricity. (In the case of prosumers, they also fulfill this role when prosumers' RES do not generate sufficient volumes of electricity). Retailers buy electricity in bulk from generators in the wholesale market and sell it to prosumers in the retail market. In current market settings, retailers are also obliged to buy any of the clients' electricity that is not traded in the P2P electricity market and still injected back to the grid. Their main objective is to *maximize profit*, while ensuring that their clients' demand for electricity is met. Retailers have access to the following information:

- Customers' demand information: This indicates how much electricity is consumed by retailers' clients over a specific period of time. This information is valuable to make any estimates about the future demand of the market.
- **Customers' supply information:** This indicates how much electricity is injected into the grid by retailers' clients over a specific period of time. Similarly, this information is valuable to make any estimates about the future demand/supply of the market.

5.1.6. Distribution System Operators (DSO)

DSOs are the operating managers (and sometimes owners) of energy distribution networks [115], operating at low and medium voltage levels. Their main objective is to *avoid congestion*.

- **Distribution grid infrastructure:** Distribution grid refers to the final stage of the electrical grid in which electricity is distributed to homes, industry, and other end-use products. Distribution is the process of reducing power to safe customer-usable levels, and delivering the electric power to the grid.
- Smart meters' inflow information: This includes the information regarding the amount and pattern of consumption at any smart meter. The more real-time this

information, the more valuable it is. Modern smart meters make it possible to read the smart meters' information in a real-time pattern. DSO is the sole actor who has access to this information through the smart meters installed at clients' premises.

- **Smart meters' outflow information:** Similar to the smart meters' inflow, this includes the information regarding the amount and pattern of electricity provisioned at any smart meter. The more real-time this information, the more valuable it is.
- Congestion information: Congestion can be defined as violations of network constraints (voltage and frequency) due to high electricity demand or excess electricity generation.

5.1.7. Transmission System Operators (TSOs)

A TSO is responsible for maintaining the transmission network, balancing the grid, and charging suppliers transmission network fees based on the electricity consumption/provision data of the suppliers' customers in the grid. A TSO's main objective is to *balance the grid*. A TSO has the following valuable resources:

- **Transmission grid infrastructure:** Electricity transmission is the bulk movement of electricity from a generating site, such as a power plant, to an electrical substation. The interconnected lines that facilitate this movement are known as a transmission grid.
- **Balancing information:** Electricity balancing encompasses all actions and processes, on all timelines, through which TSOs ensure, in a continuous way, the system frequency remains within a predefined stability range, as set forth in the Network Code on System Operation. It complies with the amount of reserves needed with respect to the required quality. This includes deficit, surplus, and reserves for any time period.
- **Congestion information:** Transmission congestion happens when scheduled market transactions (generation and load) result in power flow over a transmission element that exceeds the available capacity for that element.
- **Demand/supply prediction:** This includes all the estimates that TSOs can have based on the comprehensive information they receive for the balancing purposes.

5.1.8. Generators

Generators are the entities who generate electricity to meet the demand for electricity by consumers/prosumers. Their main objective is to *maximize profit* by trading electricity in the wholesale and/or balancing market. They have the following valuable resource:

• **Power plant:** An industrial facility for the generation of electric power. Power plants are connected to electricity grids. Their energy sources vary widely. Most of them burn fossil fuels (e.g., coal, oil, and natural gas). Cleaner energy sources of power plants include nuclear power and, increasingly, renewables (e.g., solar, wind, wave, and hydroelectric).

5.2. Opportunities for Sharing

In this section, the existing (or new) market players who would be interested in utilizing (some of the) resources of other actors, identified in the previous section, and the attainable benefit by accessing the resource are discussed. This part answers two questions: *who is interested in the resource*? and *what benefit can the interested actor gain*?

5.2.1. Prosumers

- **Renewable Electricity Sources (RES):** Other *prosumers/consumers* (or *representatives* on behalf of them), by accessing prosumers' RES, can produce renewable electricity. The owner of the produced electricity is the party with which it is shared. In a sense, rather than trading the electricity generated by the RES, prosumers can lend/rent their RES for a specific time period.
- **Batteries:** Other *prosumers/consumers* (or *representatives* on their behalf) can store electricity by accessing prosumers' batteries. The owner of the stored electricity is the party

with which the battery is shared. Renting out the RES and the battery in combination should be the most beneficial for both—the owner and the renter.

- Electric vehicles (EV): Electric vehicles can be used by other *prosumers* in idle time. Car sharing initiatives in a peer-to-peer manner are imaginable for this type of sharing. *Representatives* can represent underutilized capacity of electricity vehicles in a more efficient way on behalf of owners of EVs.
- **EV battery:** Other *prosumers* are interested in EV batteries as portable storage devices. People can, for instance, sell 2kw to users living close to their work place. Rather than feeding this energy to the grid at their home location, they can transport it with their EV and inject it to the grid at their work locations, potentially saving on grid use fees. *Representatives* can represent their clients' EV batteries in a more efficient way.
- **EV charging station:** EV charging stations can be used by other *prosumers* in idle times. This requires bringing the cars to the location of the station to charge them. Other prosumers can be interested in using the EV charging stations for charging their EVs. *Representatives* are also interested in expanding their service basket and in sharing charging stations more efficiently on behalf of the owners. *DSOs* would be interested as a means to help the congestion problem without occupying the distribution grid's capacity.
- Smart meters: People cannot share the smart meter, but it generates valuable data to be shared. Considering the frequency of access to the smart meter data, it reveals information that is highly valued by several actors. *DSOs* value this information for billing and solving congestion problems, *TSOs* for balancing, and *retailers* for customer consumption/production estimation purposes. *Representatives* can represent this information on behalf of their clients.
- Home energy management system (HMS): As is the case for the smart meter, the data generated by these systems are probably their most valuable output. *Representatives* can use these data for participating in various markets. It is unclear how one can benefit from sharing the system, unless a neighbor (*other prosumer*) could use the system's functionality as well. In a sense, the neighbor (or any user) could send the data of their assets and let the home management system of another user make intelligent decisions for them. It is yet to be seen how practical this solution is. It might well work in apartment buildings, where several flats use only one home management system.
- **Demand information:** *Representatives* are interested to have this information because it enables them to better represent their clients' demand and decide on the way to supply (purchasing from other prosumers, purchasing from grid, using the battery capacity, etc.). *Retailers* can plan their electricity provision based on forecasts based mainly on their clients demand/supply information. *DSOs* can use these data to better handle the distribution grid congestion problem. *TSOs* can better balance the grid by having this information. *Brokers* are interested in the part of this information which would be traded in peer-to-peer electricity markets through their channel.
- **Supply Information:** *Representatives* are interested in this information because it enables them to better represent their clients' supply and decide how to sell/share (selling to other prosumers, selling to the grid, using the battery capacity to store the produced electricity, etc.). *Retailers* can plan their electricity provision based on forecasts based mainly on their clients' demand/supply information. *DSOs* can use these data to better handle the distribution grid congestion problem. *TSOs* can better balance the grid by having this information. *Brokers* are interested in the part of this information which would be traded in peer-to-peer electricity markets through their channel.
- **Flexibility:** *DSOs* are interested in flexibility capacity to overcome the congestion problem. *TSOs* can use this capacity as a means to balance the electricity grid. *Representatives* can represent this capacity on behalf of clients to other interested parties.

15 of 29

Aggregators do the same type of representation for their clients in the current electricity market.

5.2.2. Representatives

As discussed in the presentation of representatives in the previous section, the type of resources representatives own is mostly not relevant to sharing. However, representatives (on behalf of prosumers) have access to the same valuable resources as prosumers and facilitate sharing. What differentiates representatives' role in sharing from prosumers is that representatives have the aggregate capacity of their clients to share. The aggregation of clients' resources in the hands of a representative gives it a service basket and leverages its position to negotiate and enter to markets that is not possible for a single prosumer to access.

- Sellers' supply information: This information is about the capacity for supply, and not necessarily all the supply capacity is traded in peer-to-peer markets. The information is useful for planning purposes. *DSOs* are interested in the supply information to have a better congestion management in the distribution grid. *TSOs* can enhance balancing planning by access to this information. *Retailers* can have plan better for their electricity demand from generators, which later they will supply in the retail market.
- **Sellers' offered price:** Access to this information helps *retailers* in their pricing. Retailers' pricing is in direct competition with the supply price in peer-to-peer trading market.
- **Buyers' demand information:** This information is about the capacity for demand, and not necessarily all the demanded amount is traded in the peer-to-peer market. The information is useful for planning purposes. *DSOs* are interested in the demand information to have a better congestion management in the distribution grid. *TSOs* can plan better for balance through access to this information. *Retailers* can plan better for their electricity demand from generators, which they will later supply in the retail market.
- **Buyers' offered price:** Access to this information helps *retailers* in electricity pricing. It indicates potential customers' willingness to pay. It is also worth mentioning that this willingness is for the peer-to-peer market. Considering that retailers nowadays are also offering green electricity to their customers, it would help retailers to tailor their offering in that product segment.
- **Clearance price:** This is the price for complementary/substitute service of *retailers*. It is expected that retailers' pricing is directly impacted by this information.
- **Total traded volume:** *Retailers* and *generators* can adjust their pricing and supply by knowing the actual traded amount of electricity in the peer-to-peer trading market. *DSOs* can better manage the congestion problem in the distribution grid. *TSOs* can better balance the grid by knowing this information.

Consideration: What defines the brokers' position in the market is the type of information that they have access to. It seems some of this information (sellers' supply and price information, and buyers' demand information and offered price) are VRIN resources that, despite the existence of other interested parties, brokers are not supposed to share.

5.2.3. Aggregators

- **Clients' supply capacity:** *TSOs* are interested in this information for prevention of congestion in the transmission grid.
- **Clients' demand capacity:** *TSOs* are interested in this information for prevention of congestion in the transmission grid.
- Clients' balancing capacity: TSOs are interested because of the use of this capacity in balancing of the grid.

5.2.4. Retailers

Customers' demand information: DSOs are interested in this information for distribution grid congestion prevention. TSOs are interested because of balancing purposes.

• **Customers' supply information:** As above.

5.2.5. Distribution System Operators (DSOs)

- Distribution grid infrastructure: *Prosumers* are interested in the distribution grid to receive their electricity (purchased in the retail or peer-to-peer market) through it. Connectivity to the grid is also necessary for receiving the balancing services which translates into the stability of the electricity stream. *Representatives* have the same dependency/interest as prosumers regarding the distribution grid. The distribution grid makes the existence of *retailers'* services meaningful. This means that they can only deliver what they sell if the client is connected to the grid.
- Smart meters' inflow information: *Retailers* are interested in this information because it reveals their customers' consumption behavior. It is valuable for pricing and planning purposes. *Representatives* are also interested in this information because they are the sellers in the peer-to-peer market. Electricity sold in the peer-to-peer market is in competition with retailers' offers. So it has a similar value for representatives. Considering the level of expertise of prosumers, the required expertise to process this information is absent in individual prosumers.
- **Smart meters' outflow information:** As above.
- **Congestion information:** *Representatives* are interested in this information because it shows where the situation is more prone to peer-to-peer trading (considering that peer-to-peer trading of electricity may cause congestion problems in the distribution grid). *Retailers* have similar interests as representatives in this information.

5.2.6. TSOs

- **Transmission grid infrastructure:** *Generators* and *prosumers* are already making use of this infrastructure by using it as a transposition means for their traded electricity.
- **Balancing information:** Active players in the balancing market are the interested parties. This includes *retailers, aggregators,* and *generators*. Other *TSOs* (neighboring countries) are interested in these data because of the balancing purposes.
- **Congestion information:** *Aggregators,* other *TSOs,* and *generators* are interested actors in this information. As a source of balancing solutions, this information is meaningful for the interested actors to find where there is a good node to offer their balancing services.
- **Demand/supply prediction:** Active players in the balancing market are the interested parties. This includes *retailers, aggregators,* and *generators*. Other *TSOs,* (neighboring) countries, are interested in these data because for balancing purposes. *DSOs* are also interested in these data because of the implications that it could have on the congestion in the distribution grid.

5.2.7. Generators

 Power plants: *Retailers* are interested in the production capacity of power plants rather than buying the electricity they produce. Retailers can keep it as a reserve for balancing purposes. This scenario makes sense if sharing the capacity is more beneficial for generators than selling the output electricity.

6. Analysis

6.1. Current vs. Future Opportunities for Sharing

This study was initially designed to explore sharing opportunities in future electricity markets. Interestingly, the findings revealed several opportunities already available in the present, which have been overlooked despite their potential for value creation and capturing for several actors. It goes without saying that these are missed opportunities that can be sources of value. This also resonates with the concept of increasing the exploitation of existing resources, which is one of the cornerstones of circularity [116]. To clarify this argument, by following the boundary conditions set for the study (i.e., a "sharing opportunity" is a situation in which at least one actor other than the owner is interested

in a resource), we built a matrix based on a combination of actors to highlight sharing opportunities between every two types of actors. We also highlight the presently missed opportunities and future opportunities for sharing. As shown in Table 2, matrix cells present a combination of actors that make up the two axes of the matrix. Each cell lists resources of the actor in the left column that are of interest to the actor in the top row. We underline the items that are currently shared. The remaining items in the list are opportunities for sharing in the future.

One of the implications of the identified sharing opportunities is that we can expect new entrants [35] (e.g., providers of sharing platforms or other supporting services) in electricity markets. Attractive sharing opportunities may attract these actors, which will also lead to facilitation of trading. Among the newly introduced actors, representatives are credible candidates to extend their activities and seize these opportunities [115]. However, privacy concerns may stand in the way of full-fledged, active participation of all prosumers [39]. As a consequence of privacy concerns, which go beyond issues of data protection, prosumers may apply a privacy/sharing calculus, weighing sharing risks against benefits [117]. In that calculus, the compounded physical privacy and data protection concerns may weigh more heavily against the benefits of sharing than data protection concerns alone would do. Those promoting sharing opportunities in electricity markets would do well to consider not only security and data protection measures but also to address the physical privacy concerns that may impede (some) prosumers from reaping the benefits.

Among the existing players in the electricity market, aggregators currently more or less undertake the combined activities of brokers and representatives [13]. A future opportunity imaginable for aggregators is to diversify their activities to offer representative and broker services. An interesting question that remains open is whether a single actor will play representative, broker, and aggregators' roles [115]. It is yet to be seen if specialization and a division of labor, meaning distinct actors undertaking separate roles [13], is the path forward, or if diversification would generate enough return for aggregators.

	Pro	Rep	Bro	Agg	Ret	DSO	TSO	Gen
Pro	Physical assets: - RES - Batteries - EV - EV charging station - Battery of EV - HMS	Physical assets: - RES - Batteries - EV - EV - EV - HMS Data: - SM info - Demand info - Supply info - Flexibility	Data: - Demand info - Supply info		Data: - SM info - Demand info - Supply info	Data: - SM info - Demand info - Supply info - Flexibility	Data: - SM info - Demand info - Supply info - Flexibility	
Rep	Physical assets: - Clients' RES - Clients' Batteries - Clients' EV - Clients' EV charging station - Clients' Battery of EV - Clients' HMS	Physical assets: - Clients' RES - Clients' Batteries - Clients' EV - Clients' EV charging station - Clients' Battery of EV - Clients' HMS	Data: - Clients' Demand info - Clients' Supply info		Data: - Clients' SM info - Clients' Demand info - Clients' Supply info	Data: - Clients' SM info - Clients' Demand info - Clients' Supply info - Clients' Flexibility	Data: - Clients' SM info - Clients' Demand info - Clients' Supply info - Clients' Flexibility	
Bro					Data: - Sellers' supply info - Sellers' offered price - Buyers' demand info - Buyers offered price - Clearance price - Total traded volume	Data: - Sellers' supply info - Buyers' demand info - Total traded volume	Data: - Sellers' supply info - Buyers' demand info - Total traded volume	Data: - Total traded volume
Agg							Data: - Clients' supply capacity - Clients' demand capacity - Clients' balancing capacity	
Ret						Data: - Customers' demand info - Customers' supply info	Data: - Customers' demand info - Customers' supply info	
DSO	Physical assets: - Distribution grid infrastructure	Physical assets: - Distribution grid infrastructure Data: - SM inflow info - SM outflow info - Congestion info			Physical assets: - Distribution grid infrastructure Data: - SM inflow info - SM outflow info - Congestion info			
TSO				Data: - Balancing info - Congestion info - Demand/supply prediction	Data: - Balancing info - Demand/supply prediction	Data: - Demand/supply prediction	Physical assets: - Transmission grid infrastructure Data: - Balancing info - Congestion info	Data: - Balancing info - Congestion info - Demand/supply prediction
Gen					Physical assets: - Power plants			

Table 2. Opportunities for sharing between different actors.

Note: Underlined items are the resources which are shared in the current market between the two involved actors.

6.2. Privacy and Data Protection Aspects of Prosumers' Sharing Opportunities

Despite opportunities to generate additional income, promote sustainable consumption, or enjoy the social activity of sharing [118], prosumers may be reluctant to share assets due to privacy concerns [114]. Such concerns may vary, from a desire to limit access to private spaces to discomfort with sharing information about daily behavioral patterns, or distrust against commercial parties whose access to sensitive data may lead to price or other market manipulation attempts. This section presents an analysis of privacy and data protection aspects of the sharing opportunities for prosumers discussed above, as well as a number of potential remedies.

First, a disambiguation between "privacy" and "(personal) data protection" is in order. A comprehensive, systematically developed typology of privacy distinguishes eight basic types: bodily, intellectual, spatial, decisional, communicational, associational, proprietary, and behavioral privacy, with informational privacy as a ninth type that overlaps, but does not coincide with, the eight basic types [94]. This non-exhaustive typology underscores the importance of protecting more than merely personal information when it comes to privacy.

The distinction between different types of privacy is also a legal one. The right to privacy was established in the Universal Declaration of Human Rights (UDHR), which states that "[n]o one shall be subjected to arbitrary interference with his privacy, family, home or correspondence, nor to attacks upon his honour and reputation". The European Convention for Human Rights (ECHR) of the Council of Europe also provides protection for the right to privacy ("respect for private life, family life, home, and correspondence") in its Article 8. This right covers a wide range of issues in the rulings of the European Court of Human Rights (ECHR): bodily integrity, wiretapping, gender identification, and unwarranted trespassing and protection against environmental nuisances, among other issues [119]. The Treaty on the Functioning of the European Union distinguishes a separate right to the protection of personal data in its Article 16. So does the European Union's Charter of Fundamental Rights, which includes both the right to respect for private and family life in Article 7 and the right to protection of personal data in Article 8. The EU's General Data Protection Regulation (GDPR) regulates this latter right, not the right to private and family life.

In the context of prosumers sharing assets in the electricity market, concerns related to physical assets are discussed here as "privacy concerns", whereas concerns related to digital assets are discussed as "data protection concerns". These labels are chosen for reasons of brevity, not to introduce a strict distinction between informational or data privacy and other privacy concerns. Our aim is to clarify that privacy concerns do not end when (personal) data are secure.

To be clear, this section does not deal with the situation in which prosumers are handling other individuals' data, i.e., the situation in which the prosumer becomes a "controller" in terms of the GDPR. What is discussed are potential personal privacy and data protection concerns of prosumers related to sharing their own assets.

6.3. Privacy Aspects of Sharing Prosumers' Physical Assets

Prosumers experience privacy concerns when they fear an infringement of personal boundaries around their bodies, intimate spaces, or personal assets. Prosumers' physical resources, as listed above, are likely to be found in or around their private homes, and possibly the homes of their families. In "collaborative consumption" schemes [118], privacy concerns about giving strangers access to zones of intimacy (which can include electric vehicles) may affect prosumers' willingness to share. Compared to data protection issues, these physical privacy concerns have been understudied, but they can weigh more heavily than concerns about the protection of personal data [117]. Among the eight basic types of privacy distinguished by [94], five are primarily related to physical aspects: bodily, spatial, behavioral, associational, and proprietary privacy. Each of these types is discussed below in relation to the sharing of physical resources of prosumers in the electricity market.

alone from actively participating in collaborative consumption schemes.
Like bodily privacy, spatial privacy, particularly the privacy of the home, is a constitutionally protected right around the world [94]. In some countries, "dwellings" or (in Poland) vehicles are also protected, while the inviolability of property, of computers, or of cell phones sometimes also falls under the protection of the private sphere. Clearly, the need to protect intimate spaces is felt worldwide. This implies that sharing of all physical assets mentioned in Section 5.1—renewable electricity sources, home batteries, electric vehicles, EV charging stations, EV batteries, home energy management systems, and smart appliances—can, to some extent, and depending on circumstances, trigger spatial privacy concerns.

privacy concerns aside in this discussion. It is, however, of no small importance to realize that related concerns may dissuade, for instance, female prosumers who live

- **Behavioral privacy** has to do with excluding others from observing one's personal actions and behaviors [120], with systemic observation being of particular concern. Visitors to a private space (when sharing renewable electricity sources, electric vehicles, home or EV batteries, EV charging stations, or smart appliances) would undoubtedly learn something of the sharing prosumer's lifestyle, actions, and personal behavior. However, systemic observation is unlikely to occur in the physical environment, but is potentially of graver concern in relation to information-sharing, or information that could be gleaned from the sharing of a home energy management system.
- Associational privacy refers to an "individuals' interests in being free to choose who they want to interact with: friends, associations, groups, and communities" [94]. Assuming that prosumers would voluntarily engage in collaborative consumption schemes, associational privacy is unlikely to be affected by the choice to share in itself. In fact, social influence plays an important role in users' sharing decisions: "The more the people in the sharers' environment encourage and support their sharing, the more frequently these users will share" [117]. Collaborative consumption therefore appears to be rather a positive expression of associational privacy preferences.
- **Proprietary privacy** is about property-based interests, meaning that property can be used to shield activity or information from others. Window curtains, locked doors, or closed bags are examples of physical property protections. Naturally, if property is shared, proprietary privacy is affected, but not necessarily infringed upon. Proprietary privacy is mostly a legal issue, in that it mostly applies to situations of unreasonable search and seizure [94], not to voluntary sharing. It is therefore less likely to be a concern specific to collaborative consumption.

For some assets, mitigation of physical privacy concerns can be achieved by introducing "safe spaces", away from intimate zones. For example, lighter smart appliances can be handed over in a public space. For less mobile physical assets, it will be of utmost importance to establish trust between sharing partners, e.g., through privacy-preserving features of sharing platforms or by introducing physical security checks, setting clear boundaries in contracts, or introducing sharing policies [121,122].

For all five types of physical privacy concerns, there are related data protection concerns. Information "leaked" about bodies, spaces, behavior, property (e.g., photos or videos taken surreptitiously), or associations (e.g., contact lists) may be as much a concern, or even more worrisome, to prosumers.

6.4. Data Protection Issues of Sharing Prosumers' Digital Resources

While informational privacy is considered to be a somewhat separate type of privacy that overlaps with the eight basic types, three of those basic types are also closely related

to the protection of data: communicational, intellectual, and decisional privacy. These types, and informational privacy, are discussed below in relation to the sharing of digital

• **Communicational privacy** is also protected in all constitutions, specifically mediated communications (or "correspondence" in the ECHR) [94] (Unmediated communication, or communication in person, is sometimes protected in the same vein, or sometimes considered as a part of physical privacy). It refers to concerns regarding conscious communication, such as sending messages on sharing platforms or online marketplaces (potential "representatives"). Communicational privacy is affected if platforms are processing communications data [122] or in case of "eavesdropping" [114].

resources of prosumers in the electricity market.

- Intellectual and decisional privacy are two sides of the same coin: intellectual privacy can be regarded as freedom from intrusion into the functioning of the mind, and decisional privacy is seen as the freedom to exercise one's mind. Prosumers sharing smart meter data, demand and supply data, and/or flexibility information with retailers or system operators may fear that their beliefs about prices and demand may be manipulated by incorrectly presented or skewed information provision by larger players in the market, affecting their ability to make effective, rational decisions in their own best interest.
- Lastly, **informational privacy** concerns around P2P sharing in the electricity market are well documented, including such risks as impersonation, data manipulation, and individual privacy breaches leaking location or trajectory data, payment information, or behavioral patterns to third parties [114,122]. Any kind of privacy also has an informational aspect. At the same time, information always relates to certain aspects of people's physical situations, which have privacy elements beyond information.

Fortunately, there is also ample discussion regarding mitigating data protection issues, partly due to the introduction of the GDPR, which requires that organizations processing personal data take extensive "technical and organizational measures" to protect such data. Technical measures can include secure authentication of parties and messages, encryption, anonymization and pseudonymization, secure channels, prosumer data aggregation, decentralized data storage, unlinkability, multiparty computation, and zero-knowledge proofs [114]. Organizational measures can include access controls, logging, contractual clauses, staff privacy training, confidentiality agreements, and binding policies, to name but a few common approaches. Figure 4 summarizes the privacy and data security aspects of sharing prosumers' physical assets as well as their data.



Figure 4. Privacy and data security aspects of sharing prosumers' physical assets as well as their data.

7. Discussion

The findings from this study suggest how prosumers' open business models, in decentralized market models in the electricity market, may be hindered by privacy concerns on sharing physical assets as well as data protection issues with sharing their data.

Our results showed several sharing opportunities in electricity markets exist which are already exploitable by prosumers for value creation and capturing in their business models. Furthermore, all the identified opportunities have a collaborative nature through resource sharing, which emphasizes the openness aspect of prosumers' business models. This is in line with previous studies' claims [46–48] on increasing use of open business models for exploring and exploiting external resources and knowledge. Furthermore, by identifying sharing opportunities between dual combination of actors it provides empirical evidence for [40,52]'s claim that open business models allow greater value capture by utilizing a firm's key asset, resource, or position not only in that organization's own operations, but also in other companies' businesses. Previous studies [29,49] mostly take a positive tone and focus on antecedents of open business models. Nevertheless, our analysis on prosumers' privacy and data protection concerns complements this strand of research by showing barriers for implementing open business models.

Somewhat surprisingly, most of the identified sharing opportunities are already exploitable. In other words, building open business models to make use of these opportunities is not hindered by legal constraints [24] or other structural changes in the electricity grid [123]. This sheds new light on previous studies that suggest the large-scale rollout of P2P, CSC, and TE models, as non-stand-alone innovations, requires several co-innovations and adoptions [13]. Furthermore, the results broaden the scope of sharing, which was mostly focused specific devices (charging stations, batteries, etc.) in previous studies [124], by putting prosumers as the core of discussion rather than a specific device.

Formerly, prosumers were considered passive actors in the electricity market because they could only inject their excess electricity into distribution grids and had no negotiation power on price setting nor on whom to sell [106,125]. As a result of the limited decisions they could take, having a business model for active prosumer participation in the market was deemed unnecessary. However, technology breakthroughs in the production of high-capacity batteries, distributed renewable resources (e.g., solar panels), ease of communication, and prevalence of smart devices (smart meters, home energy management systems, etc.), as shown under actors' resource base in our study, are expanding prosumer control and allowing them to play a more substantial active role in future electricity markets [126,127]. This shift in prosumers' capabilities necessitates better understanding of their options for value generation and capturing through their business models. A clear picture is necessary for prosumers as potential active participants in future electricity markets and for policy makers to better prioritize public resource allocation [32].

By following [128]'s guidelines for making theoretical contributions, we applied the resource-based theory of the firm in relatively new context—the electricity market—to identify resources with sharing potential held by different existing and emerging actors. This is a novel application for the resource-based theory to discover sharing opportunities [41]. Hence, our study contributes to theories of the firm, in particular to the resource-based theory [128]. Furthermore, by highlighting the privacy and data protection aspects on prosumers' sharing of physical assets and data, as potential barriers for open business models, the study contributes to open business models literature. Finally, by analyzing the differences between privacy [44] aspects of sharing physical objects and data protection [45] issues related to sharing prosumers' data in electricity markets, we contend that physical privacy aspects are too often overlooked, though they can present critical barriers to prosumer sharing.

8. Limitations and Opportunities for Future Research

This study has limitations that can be overcome in future research. Furthermore, this study raises important questions for follow-up research. First, it would be fruitful to pursue further research into the dynamics of prosumers' resource transformation in

the short run and in the long run. The intersection of open business models [29], digital transformation [129], and dynamic capabilities [130] is another area for further research. The study is partly based on speculations derived from case studies. A follow-up, quantitative study (e.g., a survey) with a large sample size [131] can validate our findings. Among the drawbacks for implementing open business models by prosumers, we focused on privacy as well as data protection concerns. Taking other perspectives (technical, legal, etc.) to study the factors which hinder applying open business models can contribute to theory and practice. Broadening the scope to cover the business models as well as privacy and security aspects of other actors is another path to follow in future studies.

It is valuable to perform an in-depth privacy analysis of identified sharing opportunities that goes beyond concerns over sharing data or the security of IT infrastructure. While there is evidence that prosumers hesitate to participate in particular parts of the sharing economy due to concerns about the privacy of their homes, bodies, or other aspects of their private lives (e.g., in schemes such as Airbnb), there has been no empirical research into prosumers' privacy-related hesitations to participate in sharing schemes for physical assets in the electricity market [132]. This may be partly due to the lack of existing schemes. Although our data collection and validation did not allow for such empirical exploration of potential concerns related to all different types of privacy, there are certainly research strategies that do allow for exploration of future attitudes, e.g., using fictional methods [133] such as "futuristic autobiographies" [134].

Last but not least, the implicit assumption in identifying opportunities for sharing in our study, which is based on scenarios for the future [135,136], is that we can extrapolate and speculate about the future using the knowledge that we have at present. Hence, the possibility of disruption as the result of breakthrough innovations [137] is overlooked in this method. These could alter the whole market. Taking into account possible sources of disruption can be another inspiration for conducting follow-up studies.

9. Conclusions

This article explores the barriers for prosumers' open business models due to their critical role in decentralized modes of electricity/flexibility trading. In particular, it puts forward privacy and data protection concerns on sharing prosumers' physical assets and data. This is an exploratory study that utilizes a multiple case study method to identify existing and emerging actors in electricity markets. Furthermore, resource-based theory is applied to identify resources of the actors under physical assets and data categories. Objectives that different actors pursue in the electricity market are identified, and for each identified resource, interested actors and their motivation are introduced. Accordingly, a matrix is built by dual combination of actors, to identify what opportunities for sharing exist between every two actors. We built on [32]'s perspective and dealt with prosumers as firms in applying the resource-based theory and in our discussions. Identified sharing opportunities, besides participation in the new market models for electricity/flexibility trading, call for developing prosumers' open business model and highlight less attended opportunities. The study contributes to theories of the firm by applying the resource-based view in a new context and to the business model literature by shedding light on barriers in applying open business models.

Author Contributions: Conceptualization, M.M., I.V.Z. and P.B.; methodology, M.M. and I.V.Z.; data curation, M.M.; writing—original draft preparation, M.M. and I.V.Z.; writing—review and editing, M.M. and I.V.Z.; visualization, M.M. and I.V.Z.; supervision, P.B.; project administration, M.M. All authors have read and agreed to the published version of the manuscript.

Funding: This work was supported in part by the Flemish Government through the FWO-SBO project SNIPPET-S007619.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: The authors would like to thank the co-authors of the earlier version of this research, which was published in the proceedings of the 16th IEEE International Conference on Distributed Computing in Sensor Systems (DCOSS).

Conflicts of Interest: The authors declare no conflict of interest.

Abbreviations

The following abbreviations are used in this manuscript:

ICTs	Information and Communication Technologies
P2P	Peer-to-Peer
CSC	Community Self-Consumption
TE	Transactive Energy
B2C	Business-to-Consumer
B2B	Business-to-Business
VRIN	Valuable, Rare, Imperfectly imitable and Not substitutable
RES	Renewable Energy Sources
EV	Electric Vehicle
HMS	Home energy Management System
DSO	Distribution System Operator
Pro	Prosumer
Rep	Representative
Br	Broker
Agg	Aggregator
Ret	Retailer
Gen	Generator
TSO	Transmission System Operator
RES	Renewable Electricity Sources
EV	Electric Vehicle
HMS	Home energy Management System
UDHR	Universal Declaration of Human Rights
ECHR	European Convention for Human Rights
ECtHR	European Court of Human Rights
GDPR	General Data Protection Regulation
DCOSS	Distributed Computing in Sensor Systems

Appendix A. Guideline Questions for Semi-Structured Interviews

- Would you please introduce yourself and your organization?
- What are your organization's roles in the electricity market?
- How do you describe the current electricity market (Who are actors and what are their roles)?
- What are the main influencers on the future of electricity market?
- How do you describe the future electricity market (Who are actors and what are their roles in next 5 to 10 years)?
- What are the constraints for P2P electricity trading?
- What are the best and worst scenarios for P2P electricity market?
- Who are the main actors and their roles in the P2P energy trading?
- Do existing trading mechanisms in electricity market need to be changed for P2P electricity trading (How should they be in order to allow for P2P trading)?
- Which non-energy-related market mechanism is applicable for P2P electricity trading?
- What roles would your company have in the future electricity market?
- What are your organization's objectives and business model in P2P electricity market?
- What problem is peer-to-peer is trying to solve?
- Which type of actors are trustworthy to have the critical roles in P2P, CSC, and TE models for electricity/flexibility trading?

References

- 1. Schellnhuber, H.; Cramer, W.; Nakicenovic, N.; Wigley, T.; Yohe, G. *Avoiding Dangerous Climate Change*; Cambridge University Press: Cambridge, UK, 2006.
- 2. United Nations Development Programme. Fighting climate change: Human solidarity in a divided world. In *Human Development Report 2007/2008;* Springer: London, UK, 2007.
- 3. Claeys, G.; Tagliapietra, S.; Zachmann, G. *How to Make the European Green Deal Work*; Bruegel Policy Contribution Issue n° 14 November 2019; Bruegel: Brussels, Belgium, 2019.
- 4. Streck, C.; Keenlyside, P.; Von Unger, M. The Paris Agreement: A new beginning. *J. Eur. Environ. Plan. Law* 2016, 13, 3–29. [CrossRef]
- Verbong, G.; Geels, F. The ongoing energy transition: Lessons from a socio-technical, multi-level analysis of the Dutch electricity system (1960–2004). *Energy Policy* 2007, 35, 1025–1037. [CrossRef]
- 6. Hoppe, T.; van Bueren, E. Guest editorial: Governing the challenges of climate change and energy transition in cities. *Energy Sustain. Soc.* **2015**, *5*, 19. [CrossRef]
- Armaroli, N.; Balzani, V. The future of energy supply: Challenges and opportunities. *Angew. Chem. Int. Ed.* 2007, 46, 52–66. [CrossRef] [PubMed]
- 8. Andoni, M.; Robu, V.; Flynn, D.; Abram, S.; Geach, D.; Jenkins, D.; McCallum, P.; Peacock, A. Blockchain technology in the energy sector: A systematic review of challenges and opportunities. *Renew. Sustain. Energy Rev.* **2019**, *100*, 143–174. [CrossRef]
- 9. Jamasb, T.; Pollitt, M. Electricity market reform in the European Union: Review of progress toward liberalization & integration. *Energy J.* **2005**, *26*, 11–41.
- 10. Lange, O. A note on innovations. Rev. Econ. Stat. 1943, 25, 19-25. [CrossRef]
- 11. LeBaron, B.; Tesfatsion, L. Modeling macroeconomies as open-ended dynamic systems of interacting agents. *Am. Econ. Rev.* 2008, 98, 246–250. [CrossRef]
- Montakhabi, M.; van der Graaf, S.; Madhusudan, A.; Sarenche, R.; Mustafa, M.A. Fostering Energy Transition in Smart Cities: DLTs for Peer-to-Peer Electricity Trading. In Proceedings of the 2021 17th International Conference on Distributed Computing in Sensor Systems (DCOSS), Pafos, Cyprus, 14–16 July 2021; pp. 466–472.
- 13. Montakhabi, M.; Vanhaverbeke, W. Ecosystem Risks of Peer-to-Peer Electricity Trading. In Proceedings of the DRUID Academy 2022, Aalborg, Denmark, 12–14 January 2022.
- Montakhabi, M.; Van Der Graaf, S.; Ballon, P.; Walravens, N.; Vanhaverbeke, W. Defining the Business Ecosystem of Peer-to-Peer Electricity Trading. In Proceedings of the 6th International Conference on New Business Models, Halmstad, Sweden, 9–11 June 2021.
- 15. Chesbrough, H.; Vanhaverbeke, W.; West, J. New Frontiers in Open Innovation; Oup Oxford: Oxford, UK, 2014.
- 16. IEA. Net Zero by 2050: A Roadmap for the Global Energy Sector; IEA: Paris, France, 2021.
- 17. Pratt, A.; Krishnamurthy, D.; Ruth, M.; Wu, H.; Lunacek, M.; Vaynshenk, P. Transactive home energy management systems: The impact of their proliferation on the electric grid. *IEEE Electrif. Mag.* **2016**, *4*, 8–14. [CrossRef]
- Montakhabi, M.; Vannieuwenhuyze, J.; Ballon, P. Expert Recommendations on Energy Trading Market Models using the AHP model. In Proceedings of the IEEE ENERGYCON 2022, Riga, Latvia, 9–12 May 2022.
- 19. Jogunola, O.; Ikpehai, A.; Anoh, K.; Adebisi, B.; Hammoudeh, M.; Son, S.Y.; Harris, G. State-of-the-art and prospects for peer-to-peer transaction-based energy system. *Energies* **2017**, *10*, 2106. [CrossRef]
- Adams, S.; Brown, D.; Cárdenas Álvarez, J.P.; Chitchyan, R.; Fell, M.J.; Hahnel, U.J.; Hojckova, K.; Johnson, C.; Klein, L.; Montakhabi, M.; et al. Social and economic value in emerging decentralized energy business models: A critical review. *Energies* 2021, 14, 7864. [CrossRef]
- Tushar, W.; Saha, T.K.; Yuen, C.; Smith, D.; Poor, H.V. Peer-to-peer trading in electricity networks: An overview. *IEEE Trans.* Smart Grid 2020, 11, 3185–3200. [CrossRef]
- Capper, T.; Gorbatcheva, A.; Mustafa, M.A.; Bahloul, M.; Schwidtal, J.M.; Chitchyan, R.; Andoni, M.; Robu, V.; Montakhabi, M.; Scott, I.; et al. A Systematic Literature Review of Peer-to-Peer, Community Self-Consumption, and Transactive Energy Market Models. SSRN 2021. [CrossRef]
- 23. Mustafa, M.A.; Cleemput, S.; Abidin, A. A local electricity trading market: Security analysis. In Proceedings of the 2016 IEEE PES Innovative Smart Grid Technologies Conference Europe (ISGT-Europe), Ljubljana, Slovenia, 9–12 October 2016; pp. 1–6.
- 24. de Almeida, L.; Cappelli, V.; Klausmann, N.; van Soest, H. Peer-to-Peer Trading and Energy Community in the Electricity Market: Analysing the Literature on Law and Regulation and Looking Ahead to Future Challenges; Technical Report; International Energy Agency: Paris, France, 2021.
- 25. Ketter, W.; Collins, J.; Saar-Tsechansky, M.; Marom, O. Information systems for a smart electricity grid: Emerging challenges and opportunities. *ACM Trans. Manag. Inf. Syst. (TMIS)* **2018**, *9*, 1–22. [CrossRef]
- 26. Xu, Z. The Electricity Market Design for Decentralized Flexibility Sources; Oxford Institute for Energy Studies: Oxford, UK, 2019.
- 27. Schneiders, A.; Fell, M.J.; Nolden, C. Peer-to-peer electricity trading and the sharing economy: Social, markets and regulatory perspectives. *Energy Sources Part B Econ. Plan. Policy* **2022**, 1–17. [CrossRef]
- 28. Adner, R. The Wide Lens: A New Strategy for Innovation; Penguin: London, UK, 2012.

- Montakhabi, M.; Radziwon, A. Doors wide shut-Towards A Framework for Open-Innovation, Business Model, Strategy and Beyond. In Proceedings of the WOIC 2021: 8th Annual World Open Innovation Conference, Eindhoven, The Netherlands, 8–10 December 2021.
- 30. Sanasi, S.; Ghezzi, A.; Cavallo, A.; Rangone, A. Making sense of the sharing economy: A business model innovation perspective. *Technol. Anal. Strateg. Manag.* **2020**, *32*, 895–909. [CrossRef]
- 31. Stake, R.E. Multiple Case Study Analysis; Guilford Press: New York, NY, USA, 2013.
- 32. Montakhabi, M.; van der Graaf, S.; Ballon, P.; Mustafa, M.A. Prosumers' Business Models in Future Electricity Markets: Peer-to-Peer, Community Self-Consumption, and Transactive Energy Models; University of Manchester: Manchester, UK, 2021.
- 33. Teece, D.J.; Kay, N. The Evolution of the Theory of the Firm; Edward Elgar Publishing: Cheltenham, UK, 2019.
- 34. Barney, J.B. The resource-based theory of the firm. *Organ. Sci.* **1996**, *7*, 469–592. [CrossRef]
- Montakhabi, M.; Zobiri, F.; Van Der Graaf, S.; Deconinck, G.; Orlando, D.; Vanhove, S.; Callaerts, R.; Ballon, P.; Mustafa, M.A. New roles in peer-to-peer electricity markets: Value network analysis. In Proceedings of the 2020 6th IEEE International Energy Conference (ENERGYCon), Gammarth, Tunisia, 28 September–1 October 2020; pp. 389–394.
- 36. Bygrave, L.A. Privacy and data protection in an international perspective. Scand. Stud. Law 2010, 56, 165–200.
- Barney, J.B. Resource-based theories of competitive advantage: A ten-year retrospective on the resource-based view. *J. Manag.* 2001, 27, 643–650. [CrossRef]
- Schwidtal, J.M.; Piccini, P.; Troncia, M.; Chitchyan, R.; Montakhabi, M.; Francis, C.; Gorbatcheva, A.; Capper, T.; Mustafa, M.A.; Andoni, M.; et al. Emerging business models in local energy markets: A systematic review of Peer-to-Peer, Community Self-Consumption, and Transactive Energy models. SSRN 2022, 4032760. [CrossRef]
- Montakhabi, M.; Van Der Graaf, S.; Ballon, P.; Mustafa, M.A. Sharing beyond peer-to-peer trading: Collaborative (open) business models as a pathway to smart circular economy in electricity markets. In Proceedings of the 2020 16th International Conference on Distributed Computing in Sensor Systems (DCOSS), Marina del Rey, CA, USA, 25–27 May 2020; pp. 482–489.
- 40. Chesbrough, H. Open Business Models: How to Thrive in the New Innovation Landscape; Harvard Business Press: Boston, MA, USA, 2006.
- 41. Nazir, M.I.; Hussain, I.; Ahmad, A.; Khan, I.; Mallik, A. System Modeling and Reliability Assessment of Microgrids: A Review. *Sustainability* **2021**, *14*, 126. [CrossRef]
- 42. Muhsen, H.; Allahham, A.; Al-Halhouli, A.; Al-Mahmodi, M.; Alkhraibat, A.; Hamdan, M. Business Model of Peer-to-Peer Energy Trading: A Review of Literature. *Sustainability* 2022, 14, 1616. [CrossRef]
- Neagu, B.C.; Ivanov, O.; Grigoras, G.; Gavrilas, M.; Istrate, D.M. New market model with social and commercial tiers for improved prosumer trading in microgrids. *Sustainability* 2020, 12, 7265. [CrossRef]
- 44. Zhang, D.; Li, Y.; Li, Y.; Shen, Z. Service Failure Risk Assessment and Service Improvement of Self-Service Electric Vehicle. *Sustainability* **2022**, *14*, 3723. [CrossRef]
- 45. Kousar, S.; Zafar, N.A.; Ali, T.; Alkhammash, E.H.; Hadjouni, M. Formal Modeling of IoT-Based Distribution Management System for Smart Grids. *Sustainability* **2022**, *14*, 4499. [CrossRef]
- West, J.; Bogers, M. Leveraging external sources of innovation: A review of research on open innovation. J. Prod. Innov. Manag. 2014, 31, 814–831. [CrossRef]
- 47. Snihur, Y.; Wiklund, J. Searching for innovation: Product, process, and business model innovations and search behavior in established firms. *Long Range Plan.* **2019**, *52*, 305–325. [CrossRef]
- 48. Aagaard, A. Sustainable Business Models; Springer: Cham, Switzerland, 2019.
- 49. Frankenberger, K.; Weiblen, T.; Gassmann, O. The antecedents of open business models: An exploratory study of incumbent firms. *R&D Manag.* 2014, 44, 173–188.
- 50. Sarker, S.; Sarker, S.; Sahaym, A.; Bjørn-Andersen, N. Exploring value cocreation in relationships between an ERP vendor and its partners: A revelatory case study. *MIS Q.* **2012**, *36*, 317–338. [CrossRef]
- 51. Pereira, D.; Leitão, J.; Devezas, T. Coopetition and co-innovation: Do manufacturing and service providers behave differently? In *Industry* 4.0; Springer: Cham, Switzerland, 2017; pp. 397–431.
- 52. Chesbrough, H.W. Why companies should have open business models. MIT Sloan Manag. Rev. 2007, 48, 22.
- 53. Urbinati, A.; Chiaroni, D.; Chiesa, V. Towards a new taxonomy of circular economy business models. *J. Clean. Prod.* 2017, 168, 487–498. [CrossRef]
- 54. MacArthur, E. Towards the circular economy. J. Ind. Ecol. 2013, 2, 23–44.
- 55. Ghisellini, P.; Cialani, C.; Ulgiati, S. A review on circular economy: The expected transition to a balanced interplay of environmental and economic systems. *J. Clean. Prod.* **2016**, *114*, 11–32. [CrossRef]
- Rosa, P.; Sassanelli, C.; Terzi, S. Towards Circular Business Models: A systematic literature review on classification frameworks and archetypes. J. Clean. Prod. 2019, 236, 117696. [CrossRef]
- 57. Stahel, W.R. The circular economy. Nature 2016, 531, 435–438. [CrossRef]
- Geissdoerfer, M.; Savaget, P.; Bocken, N.M.; Hultink, E.J. The Circular Economy—A new sustainability paradigm? J. Clean. Prod. 2017, 143, 757–768. [CrossRef]
- 59. Bocken, N.M.; De Pauw, I.; Bakker, C.; Van Der Grinten, B. Product design and business model strategies for a circular economy. J. Ind. Prod. Eng. 2016, 33, 308–320. [CrossRef]

- 60. Lüdeke-Freund, F.; Gold, S.; Bocken, N.M. A review and typology of circular economy business model patterns. *J. Ind. Ecol.* 2019, 23, 36–61. [CrossRef]
- 61. Atasu, A.; Dumas, C.; Van Wassenhove, L.N. The Circular Business Model. Pick a Strategy That Fits Your Resources and Capabilities. Harvard Business Review from the Magazine (July–August 2021). Available online: https://hbr.org/2021/07/the-circular-business-model (accessed on 14 March 2022).
- 62. Nussholz, J.L. A circular business model mapping tool for creating value from prolonged product lifetime and closed material loops. *J. Clean. Prod.* **2018**, *197*, 185–194. [CrossRef]
- 63. Botsman, R. Defining the sharing economy: What is collaborative consumption—And what isn't. Fast Co. 2015, 27, 2015.
- 64. Schwanholz, J.; Leipold, S. Sharing for a circular economy? An analysis of digital sharing platforms' principles and business models. *J. Clean. Prod.* 2020, 269, 122327. [CrossRef]
- 65. Kathan, W.; Matzler, K.; Veider, V. The sharing economy: Your business model's friend or foe? *Bus. Horiz.* **2016**, *59*, 663–672. [CrossRef]
- Despeisse, M.; Baumers, M.; Brown, P.; Charnley, F.; Ford, S.J.; Garmulewicz, A.; Knowles, S.; Minshall, T.; Mortara, L.; Reed-Tsochas, F.; et al. Unlocking value for a circular economy through 3D printing: A research agenda. *Technol. Forecast. Soc. Chang.* 2017, *115*, 75–84. [CrossRef]
- 67. Genovese, A.; Acquaye, A.A.; Figueroa, A.; Koh, S.L. Sustainable supply chain management and the transition towards a circular economy: Evidence and some applications. *Omega* 2017, *66*, 344–357. [CrossRef]
- 68. Zenger, T.R.; Felin, T.; Bigelow, L. Theories of the firm-Market boundary. Acad. Manag. Ann. 2011, 5, 89-133. [CrossRef]
- 69. Porter, M.E. Towards a dynamic theory of strategy. Strateg. Manag. J. 1991, 12, 95–117. [CrossRef]
- 70. Teece, D.J. Firm organization, industrial structure, and technological innovation. *J. Econ. Behav. Organ.* **1996**, *31*, 193–224. [CrossRef]
- Dierickx, I.; Cool, K. Asset stock accumulation and sustainability of competitive advantage. *Manag. Sci.* 1989, 35, 1504–1511. [CrossRef]
- 72. Helfat, C.E.; Finkelstein, S.; Mitchell, W.; Peteraf, M.; Singh, H.; Teece, D.; Winter, S.G. *Dynamic Capabilities: Understanding Strategic Change in Organizations*; John Wiley & Sons: Hoboken, NJ, USA, 2009.
- 73. Barney, J.B.; Hesterly, W.S. Strategic Management and Competitive Advantage: Concepts; Prentice Hall: Englewood Cliffs, NJ, USA, 2010.
- 74. Mahoney, J.T.; Pandian, J.R. The resource-based view within the conversation of strategic management. *Strateg. Manag. J.* **1992**, 13, 363–380. [CrossRef]
- 75. Penrose, E.; Penrose, E.T. *The Theory of the Growth of the Firm*; Oxford University Press: Oxford, UK, 2009.
- 76. Wernerfelt, B. From critical resources to corporate strategy. J. Gen. Manag. 1989, 14, 4–12. [CrossRef]
- 77. Eisenhardt, K.M.; Martin, J.A. Dynamic capabilities: What are they? Strateg. Manag. J. 2000, 21, 1105–1121. [CrossRef]
- 78. Wang, C.L.; Ahmed, P.K. Dynamic capabilities: A review and research agenda. Int. J. Manag. Rev. 2007, 9, 31–51. [CrossRef]
- 79. Williamson, O.E. Strategy research: Governance and competence perspectives. Strateg. Manag. J. 1999, 20, 1087–1108. [CrossRef]
- 80. Schilke, O.; Hu, S.; Helfat, C.E. Quo vadis, dynamic capabilities? A content-analytic review of the current state of knowledge and recommendations for future research. *Acad. Manag. Ann.* **2018**, *12*, 390–439. [CrossRef]
- 81. Zahra, S.A.; George, G. Absorptive capacity: A review, reconceptualization, and extension. *Acad. Manag. Rev.* 2002, 27, 185–203. [CrossRef]
- 82. Allen, D. The sharing economy. Inst. Public Aff. Rev. Q. Rev. Politics Public Aff. 2015, 67, 24-27.
- 83. Newbery, D.M. Problems of liberalising the electricity industry. Eur. Econ. Rev. 2002, 46, 919–927. [CrossRef]
- 84. Reiter, B. Theory and Methodology of Exploratory Social Science Research. 2017. Available online: https://ttu-ir.tdl.org/handle/ 2346/86610 (accessed on 14 March 2022).
- 85. Verbeke, A. International Business Strategy: Rethinking the Foundations of Global Corporate Success; Cambridge University Press: Cambridge, UK, 2013.
- Chen, C.D.; Zhao, Q.; Wang, J.L.; Huang, C.K.; Lee, N.C. Exploring sharing economy success: Resource-based view and the role of resource complementarity in business value co-creation. In Proceedings of the PACIS 2017, Langkawi, Malaysia, 16–20 July 2017.
- 87. Liu, Y.; Wu, L.; Li, J. Peer-to-peer (P2P) electricity trading in distribution systems of the future. Electr. J. 2019, 32, 2–6. [CrossRef]
- Kenyon, A.T.; Marjoribanks, T. Transforming media markets: The cases of Malaysia and Singapore. *Aust. J. Emerg. Technol. Soc.* 2007, *5*, 103–118.
- 89. Van Thiel, S. Research Methods in Public Administration and Public Management: An Introduction; Routledge: London, UK, 2014.
- 90. Kallio, H.; Pietilä, A.M.; Johnson, M.; Kangasniemi, M. Systematic methodological review: Developing a framework for a qualitative semi-structured interview guide. *J. Adv. Nurs.* **2016**, *72*, 2954–2965. [CrossRef]
- 91. Noy, C. Sampling knowledge: The hermeneutics of snowball sampling in qualitative research. *Int. J. Soc. Res. Methodol.* 2008, 11, 327–344. [CrossRef]
- 92. Glassner, A. Interactive Storytelling: Techniques for 21st Century Fiction; CRC Press: Boca Raton, FL, USA, 2017.
- 93. Eckhardt, J.T.; Shane, S.A. Opportunities and entrepreneurship. J. Manag. 2003, 29, 333–349.
- 94. Koops, B.J.; Newell, B.; Timan, T.; Škorvánek, I.; Chokrevski, T.; Maša, G. A Typology of Privacy. *Univ. Pa. J. Int. Law* 2016, 38, 483.
- 95. Wood, A.J.; Wollenberg, B.F.; Sheblé, G.B. Power Generation, Operation, and Control; John Wiley & Sons: Hoboken, NJ, USA, 2013.

- Wang, N.; Xu, W.; Xu, Z.; Shao, W. Peer-to-peer energy trading among microgrids with multidimensional willingness. *Energies* 2018, 11, 3312. [CrossRef]
- 97. Liu, W.; Qi, D.; Wen, F. Intraday residential demand response scheme based on peer-to-peer energy trading. *IEEE Trans. Ind. Inform.* **2019**, *16*, 1823–1835. [CrossRef]
- Long, C.; Wu, J.; Zhou, Y.; Jenkins, N. Peer-to-peer energy sharing through a two-stage aggregated battery control in a community Microgrid. *Appl. Energy* 2018, 226, 261–276. [CrossRef]
- 99. Pisello, A.L.; Petrozzi, A.; Castaldo, V.L.; Cotana, F. On an innovative integrated technique for energy refurbishment of historical buildings: Thermal-energy, economic and environmental analysis of a case study. *Appl. Energy* **2016**, *162*, 1313–1322. [CrossRef]
- Chakraborty, S.; Baarslag, T.; Kaisers, M. Automated peer-to-peer negotiation for energy contract settlements in residential cooperatives. *Appl. Energy* 2020, 259, 114173. [CrossRef]
- Zhang, C.; Wu, J.; Long, C.; Cheng, M. Review of existing peer-to-peer energy trading projects. *Energy Procedia* 2017, 105, 2563–2568. [CrossRef]
- 102. Sundararajan, A. Peer-to-peer businesses and the sharing (collaborative) economy: Overview, economic effects and regulatory issues. In Written Testimony for the Hearing Titled the Power of Connection: Peer to Peer Businesses; 2014. Available online: https://republicans-smallbusiness.house.gov/uploadedfiles/1-15-2014_revised_sundararajan_testimony.pdf (accessed on 3 May 2022).
- 103. Abidin, A.; Callaerts, R.; Deconinck, G.; Van Der Graaf, S.; Madhusudan, A.; Montakhabi, M.; Mustafa, M.A.; Nikova, S.; Orlando, D.; Schroers, J.; et al. Poster: SNIPPET—Secure and Privacy-Friendly Peer-to-Peer Electricity Trading. In Proceedings of the Network and Distributed System Security Symposium (NDSS 2020), San Diego, CA, USA, 23–26 February 2020.
- 104. Montakhabi, M.; Madhusudan, A.; van der Graaf, S.; Abidin, A.; Ballon, P.; Mustafa, M.A. Sharing Economy in Future Peer-to-peer Electricity Trading Markets: Security and Privacy Analysis. In Proceedings of the Workshop on Decentralized IoT Systems and Security (DISS), San Diego, CA, USA, 23 February 2020; pp. 1–6.
- Heinisch, V.; Odenberger, M.; Göransson, L.; Johnsson, F. Organizing prosumers into electricity trading communities: Costs to attain electricity transfer limitations and self-sufficiency goals. *Int. J. Energy Res.* 2019, 43, 7021–7039. [CrossRef]
- Montakhabi, M.; Van Der Graaf, S. Open Business Models' Actionability in Europe; EU Competition Policy Analysis. J. Bus. Model. 2021, 9, 29–34.
- 107. Moret, F.; Pinson, P. Energy collectives: A community and fairness based approach to future electricity markets. *IEEE Trans. Power Syst.* **2018**, *34*, 3994–4004. [CrossRef]
- Nizami, M.S.H.; Hossain, M.J.; Fernandez, E. Multiagent-Based Transactive Energy Management Systems for Residential Buildings with Distributed Energy Resources. *IEEE Trans. Ind. Inform.* 2020, 16, 1836–1847. [CrossRef]
- Lian, J.; Ren, H.; Sun, Y.; Hammerstrom, D.J. Performance Evaluation for Transactive Energy Systems Using Double-Auction Market. *IEEE Trans. Power Syst.* 2019, 34, 4128–4137. [CrossRef]
- Nguyen, H.T.; Battula, S.; Takkala, R.R.; Wang, Z.; Tesfatsion, L. An integrated transmission and distribution test system for evaluation of transactive energy designs. *Appl. Energy* 2019, 240, 666–679. [CrossRef]
- Masood, A.; Hu, J.; Xin, A.; Sayed, A.R.; Yang, G. Transactive energy for aggregated electric vehicles to reduce system peak load considering network constraints. *IEEE Access* 2020, *8*, 31519–31529. [CrossRef]
- 112. Hao, H.; Corbin, C.D.; Kalsi, K.; Pratt, R.G. Transactive control of commercial buildings for demand response. *IEEE Trans. Power Syst.* **2016**, *32*, 774–783. [CrossRef]
- Morstyn, T.; Teytelboym, A.; McCulloch, M.D. Designing decentralized markets for distribution system flexibility. *IEEE Trans. Power Syst.* 2018, 34, 2128–2139. [CrossRef]
- 114. Montakhabi, M.; Madhusudan, A.; van der Graaf, S.; Abidin, A.; Mustafa, M.A. Sharing Economy in Future Electricity Markets: Security and Privacy Analysis. In Proceedings of the Workshop on Decentralized IoT Systems and Security (DISS), in Conjunction with NDSS, San Diego, CA, USA, 23 February 2020.
- 115. D'Hauwers, R.; Van Der Bank, J.; Montakhabi, M. Trust, Transparency and Security in the Sharing Economy: What is the Government's Role? *Technol. Innov. Manag. Rev.* 2020, *10*, 6–18. [CrossRef]
- 116. Bocken, N.; Ritala, P. Six ways to build circular business models. J. Bus. Strategy 2021, 43, 184–192. [CrossRef]
- 117. Lutz, C.; Hoffmann, C.P.; Bucher, E.; Fieseler, C. The role of privacy concerns in the sharing economy. *Inf. Commun. Soc.* 2018, 21, 1472–1492. [CrossRef]
- 118. Hamari, J.; Sjöklint, M.; Ukkonen, A. The sharing economy: Why people participate in collaborative consumption. *J. Assoc. Inf. Sci. Technol.* **2016**, *67*, 2047–2059. [CrossRef]
- Sloot, B.v.d. Privacy as Personality Right: Why the ECtHR's Focus on Ulterior Interests Might Prove Indispensable in the Age of "Big Data". Utrecht J. Int. Eur. Law 2015, 31, 25–50. [CrossRef]
- 120. Finn, R.L.; Wright, D.; Friedewald, M. Seven Types of Privacy. In *European Data Protection: Coming of Age*; Gutwirth, S., Leenes, R., de Hert, P., Poullet, Y., Eds.; Springer: Dordrecht, The Netherlands, 2013; pp. 3–32. [CrossRef]
- 121. Zarifis, A.; Ingham, R.; Kroenung, J. Exploring the language of the sharing economy: Building trust and reducing privacy concern on Airbnb in German and English. *Cogent Bus. Manag.* 2019, *6*, 1666641. [CrossRef]
- 122. Yan, K.; Shen, W.; Jin, Q.; Lu, H. Emerging Privacy Issues and Solutions in Cyber-Enabled Sharing Services: From Multiple Perspectives. *IEEE Access* 2019, 7, 26031–26059. [CrossRef]
- 123. Joskow, P.L. Creating a smarter US electricity grid. J. Econ. Perspect. 2012, 26, 29–48. [CrossRef]

- Wu, C.; Kalathil, D.; Poolla, K.; Varaiya, P. Sharing electricity storage. In Proceedings of the 2016 IEEE 55th Conference on Decision and Control (CDC), Las Vegas, NV, USA, 12–14 December 2016; pp. 813–820.
- 125. Toffler, A. Powershift. Rev. Filos. 1992, 39, 175-178.
- 126. Parag, Y.; Sovacool, B.K. Electricity market design for the prosumer era. Nat. Energy 2016, 1, 1–6. [CrossRef]
- 127. Xu, Y.; Ahokangas, P.; Reuter, E. EaaS: Electricity as a service? J. Bus. Model. 2018, 6, 1–23.
- Makadok, R.; Burton, R.; Barney, J. A practical guide for making theory contributions in strategic management. *Strateg. Manag. J.* 2018, 39, 1530–1545. [CrossRef]
- 129. Dabrowska, J. The Good, the Bad, the Ugly, and the Beautiful of Digital Transformation: A Critical Multi-Level Research Agenda. *R&D Manag.* **2021**. [CrossRef]
- 130. Teece, D.J.; Pisano, G.; Shuen, A. Dynamic capabilities and strategic management. Strateg. Manag. J. 1997, 18, 509–533. [CrossRef]
- Trotter, R.T., II. Qualitative research sample design and sample size: Resolving and unresolved issues and inferential imperatives. *Prev. Med.* 2012, 55, 398–400. [CrossRef] [PubMed]
- 132. Garrido, G.M.; Sedlmeir, J.; Uludağ, Ö.; Alaoui, I.S.; Luckow, A.; Matthes, F. Revealing the landscape of privacy-enhancing technologies in the context of data markets for the IoT: A systematic literature review. *arXiv* **2021**, arXiv:2107.11905.
- 133. Rinehart, R. Fictional methods in ethnography: Believability, specks of glass, and Chekhov. Qual. Inq. 1998, 4, 200–224. [CrossRef]
- Cheon, E.; Su, N.M. Futuristic autobiographies: Weaving participant narratives to elicit values around robots. In Proceedings of the 2018 ACM/IEEE International Conference on Human-Robot Interaction, Chicago, IL, USA, 5–8 March 2018; pp. 388–397.
- 135. Drucker, P. Managing in a Time of Great Change; Routledge: London, UK, 2012.
- 136. De Jouvenel, B. The Art of Conjecture; Routledge: London, UK, 2017.
- 137. Christensen, C.; Raynor, M.E.; McDonald, R. Disruptive Innovation; Harvard Business Review: Brighton, MA, USA, 2013.